Author’s response to reviews

Title: Perspectives on child diarrhoea management and health service use among ethnic minority caregivers in Vietnam

Authors:

Thilde Rheinländer (thtr@sund.ku.dk)
Anders Dalsgaard (ad@life.ku.dk)
Flemming Konradsen (flko@sund.ku.dk)
Helle Samuelsen (H.Samuelsen@anthro.ku.dk)

Version: 5 Date: 29 August 2011

Author’s response to reviews:

The formatting requested by the production team have all been adressed.

Outstanding issue 1:
The on-line system has a bug; I can not get it to register the 2nd author no matter how many times I have tried. As indicated on the frontpage of the manuscript, the second authors name is Helle Samuelsen, h.samuelsen@anthro.ku.dk.
Please help me correct this bug!

Outstanding issue 2:
The layout of the tables (2 and 3) is causing problems: I have been asked to place the text from boxes in tables and only make horizontal lines visible. However, lines at the bottom of page now appears inside the tables, even inside the same cells. If the production team can kindly assist me in making these two tables correct (no lines at the bottum of the page when inside the same cell) I would highly appreciate.

On behalf of all authors,

Ms. Thilde Rheinländer, PhD. (Corresponding Author)
Department of International Health, Immunology and Microbiology
University of Copenhagen, Denmark
CSS, Øster Farimagsgade 5, Bld. 9
DK-1014 Copenhagen, Denmark
Phone secretary +45 35 32 76 26
Phone direct +45 23839931
Mail: thtr@sund.ku.dk